Author's response to reviews

Title: Adaptive Clinical Trial Designs for European Marketing Authorisation - A Survey of Scientific Advice Letters at the European Medicines Agency

Authors:

Amelie Elsaesser (elsaesse@uni-mainz.de)
Jan Regnstroem (Jan.Regnstrom@ema.europa.eu)
Thorsten Vetter (Thorsten.Vetter@ema.europa.eu)
Franz Koenig (Franz.koenig@meduniwien.ac.at)
Robert J Hemmings (Robert.Hemmings@mhra.gsi.gov.uk)
Martina Greco (martina.greco@ema.europa.eu)
Marisa Papaluca-Amati (Marisa.Papaluca@ema.europa.eu)
Martin Posch (martin.posch@meduniwien.ac.at)

Version: 2  Date: 25 July 2014

Author's response to reviews:

Dear Editor,

Please find enclosed the revised manuscript together with a point to point response to the referee’s comments. To facilitate the review we additionally uploaded the manuscript with track changes (as additional material - not to be published).

With kind regards,

Martin Posch